## HEST status, October 9th, 2018



- Error in SIS18\_TE\_ESR optics corrected (sign of injection septa GE01MU4I).
- All historical HEST IBHS saves converted to Paramodi import format and downloaded on HKR computer (ibhs2paramodi\_files directory).
- Reparation of particle counter in HHD DK2

  this week.
- Issues with signal in other particle counters in HAD, HTA, FRS – being investigated.
- Vacuum pump exchange in TH2 ongoing.